Author's response to reviews

Title: Implementing chlamydia screening: what do women think? A systematic review of the literature

Authors:

Natasha L Pavlin (npavlin@ozisp.com.au)
Jane M Gunn (j.gunn@unimelb.edu.au)
Rhian Parker (r.parker@unimelb.edu.au)
Christopher K Fairley (cfairley@unimelb.edu.au)
Jane Hocking (j.hocking@unimelb.edu.au)

Version: 5 Date: 30 August 2006

Author's response to reviews:

As requested by the editors the title has been changed to read "Implementing chlamydia screening: what do women think? A systematic review of the literature." Also the phrase 'The authors declare that they have no competing interests' has been added to the Competing Interests section. Minor grammatical corrections have been made to the text but the paper is otherwise unchanged.